Author’s response to reviews

Title: MAPPING, FRAMING, SHAPING: A FRAMEWORK FOR EMPIRICAL BIOETHICS RESEARCH PROJECTS

Authors:

Richard Huxtable (r.huxtable@bristol.ac.uk)
Jonathan Ives (j.ives@bristol.ac.uk)

Version: 2 Date: 07 Nov 2019

Author’s response to reviews:

Dear Nathan

Many thanks for accepting the article and for your latest suggestion, which we have gratefully taken on board. In our minor revisions here, we have used the language of “projects”, rather than a “programme”, although we do briefly mention the latter.

In the revised version, you will see that we have modified the title, so that this now refers to “projects”. The projects terminology was present in the previous version, but we have amplified this here e.g. on pages 3 and 5. We have also added a sentence on page 5, further clarifying our focus on the shape of an overall project (or, perhaps, programme of research comprising various projects or elements).

Thanks again for this helpful suggestion and we hope that these edits address these final comments.

With best wishes

Richard